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Week 08  starting 16 Feb 04

SM18 Operation SCHEDULE Meeting (Tuesdays)
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Bench status  and tests for this week : 
Test Coordinator this week:  Nicolay 164674 ;  MTM  Tech. on call Giorgio:  (16-3602) : 

Magnet coming soon:

3092, 3077,(without anticryostat) , 3078,  2050 (with anticryostat).
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.

· Other General Issues
1. TBD1 (1037) : Insulation  dipole vs. ground, good at warm, failed below 250 K. Special investigation.

2. Quench for warm up : Faulty quench performed for 1044. It has to be performed correctly. Useful to gain information (rf: 1033 with change of Iearth).

3. Case of 3030: After the first training  (Iq>12000 A) no ramp to 9 T was recorded. A second ramp to 9 T is needed in that case.

4. Scan of equipment ( do not forget to scan the gamelles,  special attention when shafts are entered during the training).

5. HV tests : megger stops during the tests. It happens with several mobile racks.

6. Truck left without key: difficult to remove shaft from D2.

	Date
	Problem
	Type
	Application/soft.
	Action/advise

	Tuesday_10_M


	TBE2: Scanning not possible for the mobile rack

TBD2: Full load-line: HWRC says that config file missing.


	MTM
	HWREC
	Entered manually. Maryline fixed the problem. 

MMP Config. files are done manually for the moment.

Guy performed manually the file.



	Tuesday_10_N


	TBD2: Launch of thermal cycle. TEMA blocked. HRWC in TEMA indicates “gamelles missing”.


	MTM
	HWREC
	The scan of the gamelles was forgotten by OP team.



	Tuesday_10_N


	MTF does not up-load files 

Oracle error =6508 

You are not identified as a valid user.
	MTM
	HWREC
	SM18 op not valid as login for HWREC in EDMS.

Operators have to use their own login to keep a trace.

	Tuesday_10_N

Wednesday_11_M Wednesday_11_A

Friday _13_M


	TBD1: HV test at cold failed: dipole vs ground failed at 320 V


	MTM
	
	Investigation at cold performed by Guy, Richard , Kevin and Stephane. Problem not in the IFS box

Magnet warm up. Disconnected and reconnected again.

	Thursday_12_N
	TBF1: No Qloc for quench number 3
	MTM
	
	Guy found that two cables from the switch to the chassis (Qloc, MM, FA) were not connected (new mobile rack).

	Wednesday_11_A

Thursday_12_M

Thursday_12_N
	TBF1: IAP @ cold : Two identical files created while triggering. The second file did not contain useful data.

Problems in Qloc compensation (TBF1), offsets in potaim cards  ( TBE2).


	MTM
	IAP
	

	Thursday_12_N
	TBC1:MM positioning not possible. CPU address error
	AB/CO
	
	Hubert/Adrian changed the CPU card.

	Thursday_12_N
	TBF1: MM positioning Over-range in one head sector
	MTM
	
	Cable connecting switch to amplifiers not properly pushed.

Guy advised to verify that the LEMO are pushed properly when this type of problem occurred. 

Change from a to e coil for the two head sectors during the night? 

	Friday_13_N
	TBC1:Using TEMA : message error receiving “ INFO from UDP socket”
	?
	
	Maryline advised to stop all the applications and restart sun machine connected to C bench..

	Friday_13_N
	TBC1: Quench for warm up. Start training then stop at 12100 A and trigger cards.

Faulty quench .No information
	MTM
	
	Request of Stephane: Quench for warm up is also important: we can gain information and help test coordinator. It should be performed correctly.

	Saturday_14_M
	TBF2 : Server crashed.
	IT?

AB/CO?
	
	Maryline came and fixed the problem.

	Saturday_14_A
	Cluster E : Access extremely slow to analyse the quenches from the control room.
	IT?

AB/CO?
	
	Analyse in local. (cluster E)

	Sunday _15_M


	TBE1: Training quenches. MIITS too high (33.9 then 35). Pick voltage at 128 V.
	MTM
	
	Check :Discharge of heaters.

Signals from V taps . Iearth

HV test OK

Stephane asks to proceed with minimum energy.

	Saturday_14_N
	TBF2, TBE1: unable to upload CDPT to MTF
	TS/CSE
	MTF
	Quit and re-launch the browser. 

If problem persists send e-mail + screen shots to EDMS support.








